Author's response to reviews

Title: Economic evaluation of Community Level Interventions for Pre-Eclampsia (CLIP) in South Asian and African countries: a study protocol

Authors:

Asif R Khowaja Mr (araza@cw.bc.ca)
Craig Mitton Dr (craig.mitton@ubc.ca)
Stirling Bryan Dr (stirling.bryan@ubc.ca)
Peter von Dadelszen Dr (pdadelszen@cw.bc.ca)

Version: 3  
Date: 27 April 2015

Author's response to reviews:

Dear Editor,

We have uploaded the copy of funding letter and an updated document of ethics approval for the primary CLIP Trial. The study protocol submitted is a trial-based economic evaluation, therefore, utilizes the trial infrastructure in all four countries.

Also, the supplement information for funding (attached) provides all necessary details, and highlights economic evaluation as objective number 2.5 on page 6.

We hope that the provided documents would be helpful for the editorial review.

Thank you!

Craig Mitton